Author's response to reviews

Title: The prevalence of nutritional anemia in pregnancy in an east Anatolian province, Turkey

Authors:

Leyla Karaoglu (leyla.karaoglu@rize.edu.tr)
Erkan Pehlivan (epehlivan@inonu.edu.tr)
Mucahit Egri (megri@gop.edu.tr)
Cihan Deprem (cdeprem@gmail.com.tr)
Gulsen Gunes (ggunes@inonu.edu.tr)
Metin F Genc (mgenc@inonu.edu.tr)
Ismail Temel (itemel@inonu.edu.tr)

Version: 4 Date: 8 June 2010

Author's response to reviews:

Major revision: Author list was checked and affiliation numbers were corrected.

Minor revision: all highlighting were removed. Authors contributions formatted into a paragraph. Visible vertical lines were removed from the tables. Reference 9 were replaced with Koksal O's book. Unit labels of MCV, MCH, etc were written in the text an in the tables. Minor changes were done in some sentences in the abstract and main text.